Author’s response to reviews

Title: Understanding preferences for type 2 diabetes mellitus self-management support through a patient-centered approach: a 2-phase mixed methods study

Authors:

Janice Lopez (JMLopez@its.jnj.com)
Bozena Katic (bkatic@patientslikeme.com)
Marcy Fitz-Randolph (mfitzrandolph@patientslikeme.com)
Richard Jackson (richard.alan.jackson@gmail.com)
Wing Chow (wchow3@its.jnj.com)
C. Mullins (dmullins@rx.umaryland.edu)

Version: 1 Date: 08 Jun 2016

Author’s response to reviews:

Editorial Requests:

-------------------

• Please ensure that your manuscript contains a Declarations section before the references, with ALL of the following subsections:
  * List of abbreviations
  * Ethics approval and consent to participate
  * Consent for publication – as you are not publishing identifying patient data, you can write "Not applicable" in this section
  * Availability of data and materials – see details below
  * Competing interests
  * Funding
  * Authors’ contributions
  * Acknowledgements.

All of the sections above can be found in our revised manuscript.

• The “Availability of data and materials” section should detail where the data supporting your findings can be found (including information found in any supplementary files). Where
applicable, please ensure that your data sets are deposited in public repositories, according to the standards (please see http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials for a list of repositories recommended for different types of data). If you do not wish to share your data, please state (in the “Availability of data and Materials” section) that data will not be shared, and explain the reason. If all the data supporting your findings is contained within the manuscript, or will be shared upon request, please state so in the “Availability of data and materials” section. Please note that identifying/confidential patient data should not be shared.

As stated in the “Availability of data and materials” section of our manuscript, the deidentified data set supporting our conclusions can be found at the Figshare repository: https://figshare.com/s/54d252dc2310745f43d2